Author’s response to reviews

Title: Virological failure reduced with HIV-serostatus disclosure, extra baseline weight and rising CD4 cells among HIV-positive adults in Northwestern Uganda

Authors:

Jonathan Izudi (jonahzd@gmail.com)
Sunday Alioni (alionisunday@yahoo.com)
Emmanuel Kerukadho (kerukadho@yahoo.co.uk)
David Nduguntse (davidndungutse@yahoo.co.uk)

Version: 4 Date: 28 Sep 2016

Author’s response to reviews:

Thank you for revising your manuscript according to our editorial requests. While we are satisfied with your revisions overall, I'm afraid we must ask that you further address one further revision point before we can hand your manuscript to our Production department:

Thank you for providing the participants' ages in ranges rather than specific values. Unfortunately, it has come to our attention that there is too much personal information in the data-set, such as tribe and religion, for us to be able to publish this data without written consent to publish having been obtained from all participants. Apologies for not having identified this issue in our previous request for revisions and for any inconvenience this may have caused you. We kindly ask that you remove the data-set from the additional files and state in your "Availability of data and materials" section that the data will not be shared in order to protect the participants' anonymity.

Response: Thank you. We have removed the data set from the additional files. Subsequently, we have stated in the availability of data and materials section that “Data will not be shared in order to protect participant’s anonymity”. Similarly, because the availability of data and materials section is associated with the supplementary files section, we have remove the latter as well.